Author's response to reviews

Title: Fulminant Myocarditis managed with Pulsatile Extracorporeal Life Support; use of Twin Pulse Life support (T-PLS(R))

Authors:

Eun Jeung Cho (jhong@cau.ac.kr)
Joonhwa Hong (joonhwa.hong@gmail.com)
Hyun Kang (roman00@naver.com)
Ju Won Choe (csicarus@hanmail.net)
Sang-Wook Kim (swivus@gmail.com)

Version: 3 Date: 11 November 2011

Author's response to reviews:

All of the following major revisions have been made
1. Consent statement is included
2. Authors contribution is included
3. Competing interests section is included
4. Numbering error of references is corrected